



Iwona Bąk 匝 and Katarzyna Cheba *匝

Department of Applied Mathematics in Economy, Faculty of Economics, West Pomeranian University of Technology, 70-310 Szczecin, Poland

* Correspondence: katarzyna.cheba@zut.edu.pl

Abstract: The main purpose of the paper was to identify the most frequently discussed directions of research on green transformation. In the article, both the significant similarities in the existing studies in this field, as well as the newly emerging topics of research, are presented. For this purpose, the authors used a systematic literature review with elements of statistical analyses. This kind of approach is not popularly used in literature review papers, as it differs from the research practices employed previously, which mostly concentrated on applying qualitative methods, alternatively supported by the analysis of the co-occurrence of keywords. In this paper, the authors decided to include selected methods of dimensional analysis in the systematic literature review, namely the log-linear and correspondence analyses. The main results of the presented analyses are a more detailed division of studies related to green transformations into groups focused on the areas more difficult to distinguish in terms of the traditionally conducted literature review.

Keywords: green transformation; green transition; green energy; systematic literature review; log-linear model; correspondence analysis

1. Introduction

The development of the global economy has resulted in the excessive exploitation of the planet's natural resources, which is not compatible with the principle of sustainable growth. The situation is deteriorating along with the increase in the world's population, which is approaching the level of 8 billion. The strategy of fast growth and neglecting the environmental issues followed by many decision makers means that ecological problems have evolved into social and economic ones. The worsening condition of the natural environment, which has attracted increasing attention since around the 1950s, has directed the attention of governments worldwide toward matters related to increasing ecological awareness and counteracting environmental threats [1]. In the context of the growing awareness and interest in the global changes in the natural environment, such as climate change and the loss of biodiversity, the concept of green transformation has emerged and is defined in the subject literature in many different ways. Some researchers see it as a consequence of societal collapse and thus treat it as a negative result [2]. According to Feol [3], the term "transformation" is often used as a mere metaphor, whereas other authors view transformation as an effective way of promoting environmental sustainability and societal prosperity [4]. In line with Scoones et al. [5], in order that humanity could exist on the Earth in a sustainable way, there is a need for numerous "green transformations", which should be both "top-down" in their nature, involving elitist alliances between state and business, as well as "bottom-up", supported by grassroots innovators and entrepreneurs, and at the same time constituting an element of the wider mobilisation of civic society.

Over the last decade, green transformation gained popularity both in the scientific literature and also in political discourse [6]; however, there is scant research explaining the meaning of this phenomenon based on diverse viewpoints and using the method of



Citation: Bak, I.; Cheba, K. Green Transformation: Applying Statistical Data Analysis to a Systematic Literature Review. *Energies* **2023**, *16*, 253. https://doi.org/10.3390/ en16010253

Academic Editor: Wen-Hsien Tsai

Received: 30 October 2022 Revised: 18 December 2022 Accepted: 22 December 2022 Published: 26 December 2022



Copyright: © 2022 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). multivariate statistical analysis in order to find connections between the terms related to green transformation. This article is a contribution to filling the research gap in this area.

The aim of the article was to identify the main directions of research on the green economy, including both the definition of the significant similarities in the existing studies in this field as well as the newly emerging topics of research. In order to realise this aim, the authors used a systematic literature review with elements of statistical analyses. The approach used by the authors differs from the research practices employed previously, which mostly concentrated on applying qualitative methods, alternatively supported by the analysis of the co-occurrence of keywords. It was also decided to include in the systematic literature review selected methods of dimensional analysis, namely the log-linear and correspondence analyses. The obtained effect was not only a map of the connections between the keywords most frequently used by authors but also a more detailed division of studies related to green transformations into groups focused on the areas more difficult to distinguish in terms of the traditionally conducted literature review.

The study consists of six sections. Following the introduction describing the research aim and the main motivations for the authors to address this subject, Section 2 presents a review of publications concerning the green economy and green transition, focusing on the main aspects discussed by their authors. Section 3 describes the methodology of the research, the data collection procedure and the description of the methods. Section 4 discusses the obtained results, while Section 5 compares the results with those obtained by other studies; Section 6 presents the conclusions.

2. Green Transformation—Research Overview

The green transformation has now become one of the more important areas of research, discussed by authors representing diverse disciplines of science. The publications in this area include both the theoretical issues linked with defining the term "green transformation", as well as present analyses of the current stage of the process of transformation. There are also some popular studies comprising the results of the literature review; however, these are usually limited to a quantitative presentation of the number of citations and number of works indexed in databases. It is far less frequently possible to find their listings of the analysed research subjects and analyses focused on, for example, examining the relations linking green transformation with other areas of research. The published papers mostly have the character of a survey, and the conducted studies in practice do not apply the methods of statistical data analysis. However, besides the knowledge, among others, of how to define green transformation or describe the stages and activities undertaken as part of the process of transition from the conventional economy to that with low or even zero emissions, one should also acquire information about the methods and models employed by scholars to expand the knowledge in this scope.

A starting point for more advanced statistical analyses is a review of the definitions of green transformation. At the same time, it is worth noting that the subject literature almost equally often uses the term "green transition", frequently treating them as interchangeable [7–9]. However, others [10] point out that the concept of green transformation refers to far more complex processes than green transition. According to Stirling [11], the term "transition" describes an ordered process realised "according to tightly disciplined technical knowledge and innovations", whereas "transformations imply more diverse, emergent and unruly political alignments, challenging incumbent structures, subject to incommensurable, tacit and embodied social knowledge and innovations".

Both these terms are defined in the subject literature in many different ways. For some, green transition means "the transition towards the achievement of the climate and environmental objectives primarily through policies and investments" [12], whilst the definitions of green transformation [5,13–15], in line with its original meaning, mostly highlight the environmental results (advantages) of the ongoing changes, the impact of the transformation saving natural resources available on the Earth, and the necessity of implementing such actions in conditions of the observed unfavourable climate change.

Thus green transformation is defined, for example, in [5], as "the restructuring of political, social, and economic systems to fit within the planetary boundaries". Newell [16] described it in a similar way, stressing that industrial societies should strive for the transformation into "a climate compatible, resource-conserving and sustainable world economic order". Some authors also focus on the very course of the transformation and the areas in which it takes place. Schmitz and Becker [17] point out that green transformation can be interpreted as "practices of radical economic, societal and institutional change that transcends sectors and levels". Table 1 presents an overview of selected definitions in the field of green transition and green transformation.

| Author (Year) | Definition |
|---|--|
| WBGU (2011) [18] | "Green transformations have the objective to be socially just, equitable, and balancing the ecological, economic, and social dimensions". |
| Scoones, I; Leach, M.; Newell, P. (2015) [5] | By prefacing the transformations with the word "green", our intention is to focus on the environmental dimensions of change, but these almost inevitably raise questions of social as well as environmental justice. |
| Lederer, M.; Wallbott, L.; Bauer, S. (2018) [19] | Green transformation is "a holistic and wide-ranging set of processes". |
| Nordic Council of Ministers (2021) [6] | Green transition is "the gradual and full transition to a fossil-fuel-free, low-carbon society". |
| Zhai, X; An, Y. (2021) [20] | "Green transformation emphasizes the change from an extensive development mode (with high input, high consumption, and high pollution) to an intensive development mode (with low input, low consumption, and zero pollution)". |

Table 1. Definition and description of green transition and green transformation.

At present, researchers also examine new aspects and/or areas that are strongly linked to the process of green transformation; among the keywords mentioned in these studies, one can find the following: security [21], including mainly energy security [22], fairness [18], understood, e.g., as a transparent and respecting diverse stages of socioeconomic development, the transition towards a low or even zero-emissions economy [23]. Many authors also define green transformation in the context of technologies developing in line with the progressing greening of the economy [20,24]. However, it is worth noting that, according to Stern and Rydge [25], the development of environment-friendly technologies is not the factor that could significantly speed up the ecological transformation; there are also many other different conditions existing both in the economy as well as in societal awareness. Pertinent examples are the recent events connected with the actual geopolitical situation and the COVID-19 pandemic, whose effect is the drastic change in the existing conditions of the functioning of not just companies but also entire societies. The recent, unexpected and continuing growth of energy prices in the coming months will undoubtedly have a significant impact on changes in consumer behaviour. For this reason, the recently promoted technological solutions regarding the use of electricity-powered vehicles are becoming far less attractive commercially.

One should also note that, over the last decade, the concept of green transformation has been examined in many different studies [3,7,26–31]. The review of the literature available in the Web of Science (WoS) database suggests that in the period 2001–2022 (as of the end of September 2022), there were 871 indexed publications that referred to, in their title, abstract and/or keywords to the terms "green transformation" (524 publications) or "green transition" (382) (Table 2).

| Searched Terms | No. of Publications | |
|--|---------------------|--|
| "green transformation*" | 524 | |
| "green transition*" | 382 | |
| "green transformation*" OR "green transition*" | 871 | |

Table 2. Number of papers indexed in the WoS databases, including the indicated keywords.

There has been a systematic increase in the number of publications in the analysed period starting already from 2017 when 51 such works were published. In 2021, the WoS database indexed 191 publications, and a further 270 appeared by the end of September 2022. In line with the growing number of publications, one could also observe an increase in the number of their citations, from 280 in 2017 to 2535 in 2021 and up to 2970 in the first three quarters of 2022 (Figure 1).



Figure 1. The evolution of publications and citations by year.

Despite the growing number of publications and their citations, the assembled database is not very impressive, which only confirms the limited exploration of issues in this area of research. The emergence of the literature devoted to green transformation (or green transition) should be viewed in the context of the growing knowledge and interest in global environmental change, such as climate change and loss of biodiversity. The most recent and widely publicised political agreements, for example, the one signed in Paris, and the declared aims of sustainable development (SDG) [32], have led to an ever greater interest in transformation on the part of society and academia. The growing interest in the analysed research subject in recent times is also a direct result of the already-mentioned current geopolitical situation and the energy crisis being its consequences.

The publications indexed in the WoS database mainly refer to all the issues corresponding with the scope of environmental science ecology (402 publications), science technology (222), business economics (179), engineering (118) and energy fuels (112). They were published mostly in such journals as *Sustainability* (84), *Journal of Cleaner Production* (36), *Environmental Science and Pollution Research* (34), *International Journal of Environmental Research and Public Health* (23) and *Energies* (21). The majority of their authors come from China (429 publications), the United Kingdom (78), Denmark (70), Italy (61), the USA (57) and Germany (51). Table 3 presents the bibliographical details of the top ten most cited articles.

These are mainly publications of a general nature, corresponding with the scope of theoretical considerations dedicated to green transformation. Singh et al. [33] examined the way in which the green management of human resources impacts the relations between green transformative leadership, green innovation and environmental effectiveness. The authors mostly used data from a questionnaire survey conducted in 309 small and medium enterprises in the production sector and verified the posed hypotheses by employing the covariate modelling of structural equations (SEM).

| Title | Journal | Publications | Total Citations |
|--|--|--------------|--------------------|
| Green innovation and environmental performance: The role of green transformational leadership and green human resource management | Technological Forecasting and Social Change, 150, 119762 | [33] | 325 |
| The determinants of green product development performance: Green dynamic capabilities, green transformational leadership, and green creativity | Journal of Business Ethics, 116(1), 107–119 | [34] | 254 |
| Aggregation-and leaching-resistant, reusable, and multifunctional Pd@ CeO2 as a robust nanocatalyst achieved by a hollow core-shell strategy | Chemistry of Materials, 25(9), 1979–1988 | [35] | 208 |
| Sustainability transitions: A political coalition perspective | Research Policy, 43(2), 278–283 | [36] | 195 |
| Green, Transition-Metal-Free Aerobic Oxidation of Alcohols Using a Highly Durable Supported Organocatalyst | Angewandte Chemie International Edition, 46(38), 7210–7213 | [37] | 174 |
| The politics of green transformations | Routledge | [5] | 166 |
| Sensemaking and sustainable practicing: functional affordances of information systems in green transformations | MIS quarterly, 1275–1299 | [38] | 164 |
| Promoting sustainability of manufacturing industry through the lean energy-saving and emission-reduction strategy | Science of the Total Environment, 665, 23–32 | [39] | 142 |
| How does environmental regulation promote technological innovations in the industrial sector? Evidence from Chinese provincial panel data | Energy Policy, 139, 111310 | [40] | 129 |
| Effect of green transformational leadership on green creativity: A study of tourist hotels | Tourism Management, 57, 118–127 | [41] | 123 |

Table 3. The top ten most cited papers.

In turn, Chen and Chang [34] discussed the determinants of effectiveness in the development of ecological products. To this end, they conducted a questionnaire survey among the following: a) 12 CEOs, managers of environmental, marketing, manufacturing or R&D departments, and leaders and members of green product development projects in different electronics companies in Taiwan; b) 254 businesses. Similarly to Singh et al. [33], the study used SEM to verify the posed hypotheses.

Hess [36] examined the influence of political conditions on the speed of the process of green transformation in the United States, using mainly literature studies and a case study. The conducted research suggests that in industrialised countries, there are tensions between the need to change to low-emission energy sources and the desire to satisfy the fast increase in demand for energy with the continued exploitation of fossil fuels.

A case study was also used by Seidel et al. [38] to describe the implementation of environment-friendly business practices by a global supplier of operational software. The authors of that publication showed, among others, the way in which information systems can contribute to creating environmentally sustainable organisations.

The study by Cai et al. [39] presented a review of the current practices and the limitations of the applied energy-saving strategies and the strategies for reducing emissions (ESER) in the manufacturing industry. Based on the case study, the authors outlined a new concept of the strategy aimed at improving energy efficiency and reducing waste emissions.

In the more advanced research presented by Ouyang et al. [40], the authors used a two-directional panel model with permanent effects to analyse the impact of environmental regulations on technological innovations; the study showed the existence of a U-shaped relationship between them. In the short-term, environmental regulations have a negative impact on the development of research and innovation capabilities of the industrial sector in China; however, along with the expanding environmental regulations, companies seek to reduce the costs of controlling pollution via the improvement of their capability for technological innovation.

In all these studies, the conducted literature review had the character of synthesis based on selected works published in the analysed scope, following the subjective choice of the authors. However, despite a fairly large number of the analysed works, none of those attempted to provide a comprehensive view of the issues linked with green transformation or green transition. The authors mostly focused on the presentation of their research results, which usually described individual strategies applied by the companies under study, for example, as a case study. The fragmentary character of the existing research on green transformation (or transition) was also confirmed by the results obtained from more advanced analyses of the contents of the publications on this subject, the outcomes of which are presented in the following sections of this paper.

3. Research Methodology

3.1. Stages of the Applied Research Procedure

To analyse the content of publications on the green transformation or green transitions, the authors employed a three-step research procedure whose stages are presented in Figure 2.

The first stage, based on the created database of 871 publications indexed in the WoS, identified keywords used in them to describe green transformation or green transition and their connections with other keywords describing these terms. The study used VOSviewer version 1.6.14, which allowed for the distinction of clusters containing the keywords most frequently creating such connections. In this stage of the research, publications presenting these issues using different types of models were sought. The results of the first stage were also visualised in a map showing the links between the terminology of green transformation and green transition models with other keywords identified in the conducted study (Figure 3).

The next stages of searching for relations between the selected keywords describing the results of quantitative or qualitative research presented in the analysed studies employed the methods of multivariate statistical analysis, namely the log-linear and correspondence analyses.

3.2. Statistical Methods

The methodology of a systematic literature review, through the use of a repetitive and precisely described research procedure, allows for searching, selecting and evaluating all the available scientific evidence in a given field of research [41]. Analysis of periodical publications, above all in the form of scientific articles, provides information on the current state of research, allows for posing research questions correctly, proposing hypotheses and then helping with the interpretation of the obtained empirical results [42].

Conducting a literature review constitutes an important part of every research project [43–48], helps to understand a given issue and identifies knowledge gaps that will have to be addressed in subsequent studies. Based on the review, the authors constructed a base of academic publications containing the appropriate keywords, which in turn led to the creation of variables used in the later stages of the research. The log-linear analysis allows for a detailed assessment of dependencies between any number of non-metric variables recorded in a multivariate contingency table, as well as including interactions occurring



among them. It is used successfully in many fields of science, in particular in medicine [49–51], demographic studies [52,53] and economics [54–57].

Figure 2. Research procedure chart, where * denotes the possibility of different inflectional forms of a word.



Figure 3. Map of links among keywords characterised by most frequent co-occurrence.

In the log-linear model, it is assumed that the natural logarithm of the numerosity expectation value per cell in the independencies table is a linear function of factors. The generalised form of the model is as follows:

$$ln(\hat{n}^{T...}) = n + \sum \lambda_i^T + \ldots + \sum \lambda_{ij}^{TS} + \ldots + \sum \lambda_{ijk}^{TSQ} + \ldots$$
(1)

where $\hat{n}^{T...}$ is vector $[\hat{n}_i^T \hat{n}_{ij}^{TS} ...]$, and \hat{n}_i^T is the expected numerosity of *i*-th level of factor *T*, \hat{n}_{ij}^{TS} is the expected numerosity regarding interactions of *i*-th level of factor *T* and *j*-th level of factor *S*, etc.; *n* is the mean of natural logarithms of all observed numerosities calculated on the basis of formula

$$n = \frac{1}{n} \sum_{T} \sum_{i} \ln(n_i^T), \tag{2}$$

where \sum_{T} is summation after all the factors; \sum_{i} is summation after all the levels of the factors; λ_{i}^{T} is the indicator of *i*-th level of factor *T*; λ_{ij}^{TS} is the indicator of the interaction of the second order of *i*-th level of factor *T* and *j*-th level of factor *S*; λ_{ijk}^{TSQ} is the indicator of the interaction of the interaction of the interaction of the interaction of factor *G*, etc.

The measure of the model's fit to the results are the statistics of Pearson's χ^2 and χ^2 of the greatest reliability.

In order to provide the specification of a log-linear model, it is necessary to define the order of variable interactions, namely which interactions of a given order (among which variables) should be included in the model. To this end, one should carry out analyses of partial and boundary correlations. The former informs whether the respective interaction is statistically relevant when all the factors of the same level are already within the model, whereas the latter signals whether the respective correlation has an impact when the model does not yet have any interactions of the same order [49].

Correspondence analysis is often used for qualitative variables [58–61]. The starting point in this method is to record a complex contingency table, including numerosity of the occurrence of the individual categories of variables assumed to describe *n* objects; Burt's matrix is a very frequently used way of recording the values [62]. In the first stage, a complex matrix of markers **Z** is constructed, containing blocks (submatrices) corresponding to the following variables: $\mathbf{Z} = [\mathbf{Z}_1, \dots, \mathbf{Z}_Q]$, where Q stands for the number of characteristics. The elements of this matrix only take on values of 0 and 1, depending on whether the given object does or does not have a distinct category of a variable.

Burt's matrix is given by $\mathbf{B} = \mathbf{Z}^T \mathbf{Z}$. This provides a symmetrical block matrix, in which on the main diagonal, there are placed diagonal matrices containing the number of the occurrence of the characteristics, and outside the main diagonal, there are contingency tables for each pair of the studied variables. The total number for each submatrix equals the numerosity of the examined units *n*, while the total number for Burt's matrix amount to $n \cdot Q^2$. Since Burt's matrix is symmetrical ($b_{ij} = b_{ji}$), then the boundary numbers for lines and columns are identical and calculated as follows [63–65]:

$$\sum_{j=1}^{J} b_{ij} = b_{i.} = b_{.j=Q} = Q \cdot b_{ii}, \tag{3}$$

where b_{ij} is the elements of Burt's matrix, and *J* is the total number of categories of all the characteristics.

Marginal frequencies of lines and columns are equal and amount to

$$p_{i\bullet} = \frac{Q \cdot b_{ii}}{n \cdot Q^2} \tag{4}$$

Values $p_{i\bullet}$ are the elements of a diagonal matrix of boundary frequencies of the lines and hence also the columns; they are also the components of the vector of boundary frequencies **r**. The matrix of the observed frequencies is calculated as

$$P = \frac{1}{n \cdot Q^2} B \tag{5}$$

Matrix **B** is symmetrical and subject to decomposition according to its own values

$$A = D_r^{-1/2} \left(P - rr^T \right) D_r^{-1/2} = U \Gamma_B^2 U^T$$
(6)

where **U** is the matrix of vectors of matrix A, Γ_B^2 is the diagonal matrix containing squares of peculiar values $\gamma_{B,k}^2$ (k = 1, 2, ..., K; $K = \sum_{q=1}^{Q} (J_q - 1)$ of matrix **A**, J_q is the number of categories of characteristic **q** and **D**_r is the diagonal matrix of the observed boundary frequencies of the lines.

As $\Gamma_B^2 = \Lambda_B$, where $\lambda_{B,k}$, are the own values of matrix **A**, there follows decomposition of matrix **A** according to own values, and only one matrix contains the coordinates of the characteristics

$$\mathbf{F} = D_r^{-1/2} U \Gamma_B \tag{7}$$

The size of the actual space of the coexistence of answers to the questions is calculated based on the formula

$$K = \sum_{q=1}^{Q} (J_q - 1).$$
(8)

According to Greenacre's criterion, the best choice is the dimension of projecting categories of variables in which the own values follow the condition $\lambda_{B,k} > \frac{1}{Q}$.

Greenacre added to the complex criterion of choosing the relevant own values $(\lambda_{B,k} > \frac{1}{Q})$ the way to "improve" the results of the analysis of variables given in the form of Burt's matrix [66–68]:

$$\widetilde{\lambda}_{k} = \left(\frac{Q}{q-1}\right)^{2} \cdot \left(\sqrt{\lambda_{B,k}} - \frac{1}{Q}\right)^{2}$$
(9)

where *Q* is the number of variables, and $\lambda_{B,k}$ is the *k*-th own value.

When using the correction of Greenacre, new coordinates for the categories of characteristics are set according to the formula

$$\tilde{\mathbf{F}} = \mathbf{F}^* \cdot \mathbf{\Gamma}^{-1} \cdot \tilde{\boldsymbol{\Lambda}} \tag{10}$$

where **F** is the matrix of new coordinates for variable categories, **F**^{*} is the matrix of the primary values of coordinates for variable categories, Γ^{-1} is the diagonal reverse matrix of the peculiar values and $\tilde{\Lambda}$ is the diagonal matrix of the modified own values.

4. Study Results

4.1. The Results of Systematic Literature Review

The outcome of the analyses carried out in the first stage of the study was the classification of keywords identified in the 871 collected works into clusters containing terms commonly used by their authors to describe the analysed phenomenon. Thanks to the VOSviewer software version 1.6.14, five clusters were identified using the following expressions:

- 1. Cluster 1: China, competitiveness, countries, data envelopment analysis, economic growth, efficiency, emissions, empirical evidence, energy efficiency, energy consumption, environmental regulation, industrial green transformation, pollution, productivity growth, slacks-based measure and technological innovation.
- 2. Cluster 2: carbon emissions, CO2 emissions, consumption, drivers, economy, electricity, energy, energy consumption, governance, green transformation, green transition,

energy transformation, governance, model, simulation, system, technologies, water and wind.

- 3. Cluster 3: climate change, energy policy, transition, energy transition, green, green economy, industry, innovation, management, policy, politics, supply chain, sustainability, sustainable development, systems, technology and transformation.
- Cluster 4: climate change, COVID-19, environment, green innovation, growth, investment, perspective, policies, political economy, renewable energy, research and development.
- 5. Cluster 5: energy transfer and transition.

The results obtained in this stage were also visualised in a map presenting the connections of the terms green transformation and green transition with other keywords identified in the examined publications (Figure 3).

By taking into consideration the connections occurring among the keywords, it is possible to indicate a few recurrent elements characteristic of the research presented in the analysed publications concerning green transformation and green transition (Table 4).

Table 4. Characteristics typical of research on green transformation (or transition) in the analysed publications.

| Characteristic | Examples of Publications |
|--|---------------------------------|
| Studies describing the process of green transformation on country level (regarding economies of countries in the world) | [15,66–70] |
| Studies describing the process of green transformation on regional level (regarding the world's regions) | [69,71–74] |
| Studies describing the process of green transformation on the level of individual sectors/branches of industry: production, transport, energy, etc. | [75–79] |
| Studies describing the process of green transformation on the level of individual companies (qualitative research) or their groups (quantitative or qualitative research) | [80-84] |
| Studies presenting research on green transformation in systemic approach as an element of a greater whole linked with the current socio-economic realisations globally | [8,85–88] |
| Studies regarding issues linked with sustainable development, including economic, social and environmental | [86–92] |
| Studies presenting examples of technologies supporting the process of green transformation | [10,93–96] |
| Studies presenting examples of innovations supporting the process of green transformation | [97–100] |
| Studies addressing the issues linked with the policy of green transformation | [101–104] |
| Studies addressing the issues linked with green transformation, mainly in the context of changes related to abandoning traditional sources of energy, mostly coal, in favour of other green sources of energy | [105–107] |
| Studies presenting the results of modelling the process of green transformation | [108–113] |

4.2. The Results of the Log-Linear Models

In the second stage, the authors established a binary dependent variable defining the occurrence of the models of green transformation, as well as dependent binary variables, identified based on the analysis of the keywords selected for the study of publications. These variables show whether the examined academic publications include the issues

selected for this study. Next, a log-linear model was constructed, enabling the identification

of connections and interactions among the variables. Bearing in mind the aim of this article, the authors distinguished a dichotomous dependent variable (*Y*), describing the occurrence of green transformation models (1—yes, 0—no). The following variables were included in the set of independent variables:

 X_1 —research on the level of a country (1, 0);

 X_2 —technology (1, 0);

 X_3 —economy (1, 0);

 X_4 —environment (1, 0);

 X_5 —research on the level of a region (1, 0);

 X_6 —coal (1, 0);

 X_7 —systemic approach to green transformation (1, 0);

 X_8 —industry (1, 0);

 X_9 —companies (1, 0);

 X_{10} —politics (1, 0);

 X_{11} —social aspects (1, 0);

 X_{12} —innovation (1, 0).

In the first step, the study examined which variables demonstrate significant relations with a dependent variable, using the independence test χ^2 . Table 5 shows the values of testing statistics χ^2 and test probability *p*.

Table 5. Value of statistics χ^2 and test probability *p*.

| Variable | Value of Statistic χ^2 | Value of Test Probability <i>p</i> |
|----------|-----------------------------|------------------------------------|
| X_1 | 16.2181 | 0.0001 |
| X_2 | 10.2082 | 0.0014 |
| X_3 | 3.5961 | 0.0479 |
| X_4 | 4.8745 | 0.0273 |
| X_5 | 15.6281 | 0.0001 |
| X_6 | 5.0912 | 0.0241 |
| X_7 | 0.5540 | 0.4567 |
| X_8 | 0.3649 | 0.5458 |
| X_9 | 0.2306 | 0.6311 |
| X_{10} | 2.2785 | 0.1312 |
| X_{11} | 0.5394 | 0.4627 |
| X_{12} | 0.1416 | 0.7068 |

Based on the calculations, it can be stated that the variable describing the occurrence of green transformation models is not linked to the following variables: a systemic approach to green transformation (X_7), industry (X_8), enterprises (X_9), policy (X_{10}), social aspects (X_{11}) and innovation(X_{12}), which consequently were excluded from further research, whereas all the other variables (X_1-X_6) apart from variable Y qualified into the log-linear model. In order to provide the specification of the model, the study defined the order of interaction. The results of the tests of all the interactions are presented in Table 6, suggesting that the model that is to be analysed should include, besides the main factors, also interactions of, at most, second order. This is indicated by the values of test probabilities *p* corresponding to the values of test statistics χ^2 with the highest probability and Pearson's χ^2 .

In order to assess which interactions should be included in the log-linear model, the authors used partial and boundary tests (Table 7). Due to the significant number of interactions possible for six factors, it was decided to present the interactions of the second order at most because none of the interactions of higher orders proved to be statistically relevant in terms of the partial and boundary tests. The interactions marked in bold in Table 6 are those for which partial and boundary dependencies are relevant.

| Degree of Interaction | Degree of Freedom | Value χ^2 NW | Probability p | Pearson's χ^2 Value | Probability p |
|--------------------------|----------------------|-------------------|---------------|--------------------------|---------------|
| 1 | 7 | 183.9112 | 0.0000 | 338.8615 | 0.0000 |
| 2 | 21 | 121.1573 | 0.0000 | 213.3733 | 0.0000 |
| 3 | 35 | 28.8885 | 0.7571 | 33.2661 | 0.5520 |
| 4 | 35 | 26.7474 | 0.8402 | 24.9619 | 0.8956 |
| 5 | 21 | 8.9109 | 0.9899 | 9.1476 | 0.9880 |
| 6 | 7 | 3.9952 | 0.7803 | 4.2614 | 0.7492 |
| 7 | 1 | 0.3428 | 0.5582 | 0.3439 | 0.5576 |

Table 6. The results of variable interaction tests.

Table 7. Excerpts from the results of partial and boundary tests between variables: Y, X_1 , X_2 , X_3 X_4 , X_5 , X_6 .

| Factors | Degree of Freedom | Partial Relationship χ^2 | The <i>p</i> -Value in Partial Relationship | Boundary Relationship χ^2 | The <i>p</i> -Value in Partial Relationship |
|------------------|----------------------|----------------------------------|---|-----------------------------------|---|
| Ŷ | 1 | 11.5732 | 0.0007 | 11.5732 | 0.0007 |
| X_1 | 1 | 5.0497 | 0.0246 | 5.0497 | 0.0246 |
| X2 | 1 | 11.5732 | 0.0007 | 11.5732 | 0.0007 |
| X3 | 1 | 3.4608 | 0.0497 | 3.4608 | 0.0497 |
| X_4 | 1 | 6.0358 | 0.0140 | 6.0358 | 0.0140 |
| X_5 | 1 | 99.9059 | 0.0000 | 99.9059 | 0.0000 |
| X ₆ | 1 | 49.3135 | 0.0000 | 49.3135 | 0.0000 |
| YX_1 | 1 | 5.9548 | 0.0147 | 13.7561 | 0.0002 |
| ΥX ₂ | 1 | 5.4859 | 0.0192 | 8.9260 | 0.0028 |
| ΥX ₃ | 1 | 0.0338 | 0.8542 | 3.0132 | 0.0826 |
| YX_4 | 1 | 0.8958 | 0.3439 | 3.5874 | 0.0582 |
| YX_5 | 1 | 7.5624 | 0.0060 | 13.4593 | 0.0002 |
| ΥX ₆ | 1 | 1.7404 | 0.1871 | 5.0030 | 0.0253 |
| $X_{2 \times 2}$ | 1 | 0.7855 | 0.3755 | 3.9828 | 0.0460 |
| X_1X_3 | 1 | 8.6852 | 0.0032 | 17.6010 | 0.0000 |
| X_1X_4 | 1 | 4.0025 | 0.0454 | 9.6944 | 0.0018 |
| X_1X_5 | 1 | 9.5992 | 0.0019 | 17.7962 | 0.0000 |
| X_2X_6 | 1 | 0.7717 | 0.3797 | 4.4377 | 0.0352 |
| X_2X_3 | 1 | 2.2114 | 0.1370 | 4.6715 | 0.0307 |
| X_2X_4 | 1 | 0.0022 | 0.9630 | 1.0709 | 0.3007 |
| $X_{2}X_{5}$ | 1 | 0.1866 | 0.6658 | 2.1481 | 0.1427 |
| X_2X_6 | 1 | 2.0430 | 0.1529 | 4.0412 | 0.0444 |
| X_3X_4 | 1 | 8.1640 | 0.0043 | 13.1781 | 0.0003 |
| $X_{3}X_{5}$ | 1 | 4.4063 | 0.0358 | 9.4064 | 0.0022 |
| X_3X_6 | 1 | 0.2236 | 0.6363 | 2.8827 | 0.0895 |
| X_4X_5 | 1 | 0.2867 | 0.5923 | 0.6307 | 0.4271 |
| X_4X_6 | 1 | 11.3229 | 0.0008 | 15.1487 | 0.0001 |
| X_5X_6 | 1 | 0.3207 | 0.5712 | 1.8738 | 0.1710 |

The results of the tests presented in Table 7 indicate that the model should include six main factors, as well as interactions of the second order, which occur between dependent variable *Y* and the independent variables. Moreover, the model should also contain the result representing all the interactions between the independent variables, which will help to avoid decreasing the degree of fit. Eventually, the study included dependencies between the occurrence of the models of green transformation and the following:

- Research at the national level (*X*₁);
- Technology (X₂);

Research at the regional level (X₅).
 The log-linear model takes on the following form:

$$n(\hat{n}_{(ijklmno)}) = n + \lambda_i^{Y} + \lambda_j^{X_1} + \lambda_k^{X_2} + \lambda_l^{X_3} + \lambda_m^{X_4} + \lambda_n^{X_5} + \lambda_o^{X_6} + \lambda_{ij}^{YX_1} + \lambda_{ik}^{YX_2} + \lambda_{in}^{YX_5} + \lambda_{jklmno}^{X_1X_2X_3X_4X_5X_6}$$

The estimated model is well-fitted to the empirical data, as shown by the values of statistics χ^2 of the highest probability, which amounts to 75.637 (p = 0.994), and the statistics of Pearson's χ^2 which equal 80.095 (p = 0.969). The goodness-of-fit of the model is confirmed in Figure 4, which shows the shaping of the numerosity of those observed versus the fitted ones. The results presented in Table 7 clearly show that the models of green transformation, apart from the main factors, are also affected by interactions of second order for the following pairs of variables:

- Country and economy (*X*₁*X*₃);
- Country and environment (*X*₁*X*₄);
- Country and region (*X*₁*X*₅);
- Economy and environment (*X*₃*X*₄);
- Economy and region (*X*₃*X*₅);
- Environment and coal (*X*₄*X*₆).



Figure 4. Distribution graph of observed frequencies versus fitted frequencies.

This also suggests that studies concerning models of green transformation are conducted at both national and regional levels and are usually linked to the economy and environment.

4.3. The Results of the Correspondence Analysis

The next stage of the study involved creating six categorising variables connected with the area of the conducted research and the issues addressed in scientific publications. They regarded sustainable development, technology and innovations applied in industry and in enterprises, policy and innovation, and also models and systems of green transformation. The study included the following variables and their categories:

- A-area (1–China, 2–Europe, 3–others);
- SD-sustainable development concerning three aspects: economic, social and environmental (1–no aspect of SD was considered; 2–at least two aspects of SD were considered; 3–all three aspects of SD were considered);

- TIE-technologies in industry and enterprises (1-technologies only in industry; 2-technologies only in enterprises; 3-technologies in industry and in enterprises);
- IIE-innovation in industry and in enterprises (1-innovation only in industry; 2-innovation only in enterprises; 3-innovation in industry and in enterprises);
- PI-policy and innovation (1-no reference to both categories; 2-including only one category; 3-including both categories together);
- MS-models and systems of green transformation (1-no reference to both categories; 2-including only one category; 3-including both categories together).

The first step involved creating a Burt's matrix sized 18×18 (number of categories accepted for the study). The size of the real space of the coexistence of variables calculated based on formula 8 amounted to 12. Next, it was checked to what degree the own values of spaces of a smaller size explain the total inertia ($\lambda = 2.0000$). By using Greenacre's criterion, it was found that inertias for *K* assumed at the most 6 because only the main inertias were higher than $\frac{1}{Q} = \frac{1}{8} = 0.1667$ were considered relevant for the study. For such dimensions, the authors set a share of inertia of a chosen size (λ_k) in total inertia (λ), designated as τ_k . In addition, a diagram showing its own values was prepared (Figure 5), and by using the "elbow" criterion, it was decided that the space of the presentation of the coexistence of the inertia in this space equals 4.43%, and after the modification of the own values, according to Greenacre, 66.94% (Table 8).



Figure 5. Eigenvalue—"elbow" criterion.

Table 8. Singular values and eigenvalues together with the degree of explanation of the total inertia in the original and modified versions.

| К | Singular Values γ_k | Eigen Values λ_k | λ_k/λ | $	au_k$ | $	ilde{\lambda}_k$ | $	ilde{\lambda}_k/	ilde{\lambda}$ | $	ilde{	au}_k$ |
|---|----------------------------|--------------------------|---------------------|---------|----------------------------------|-----------------------------------|----------------|
| 1 | 0.594550 | 0.353489 | 17.67446 | 17.6745 | 0.2636 | 0.3043 | 0.3043 |
| 2 | 0.526800 | 0.277518 | 13.87589 | 31.5503 | 0.1868 | 0.2156 | 0.5199 |
| 3 | 0.466530 | 0.217650 | 10.88251 | 42.4329 | 0.1295 | 0.1495 | 0.6694 |
| 4 | 0.438582 | 0.192354 | 9.61769 | 52.0505 | 0.1065 | 0.1229 | 0.7923 |
| 5 | 0.419355 | 0.175859 | 8.79294 | 60.8435 | 0.0919 | 0.1061 | 0.8984 |
| 6 | 0.413874 | 0.171291 | 8.56457 | 69.4081 | 0.0880 | 0.1016 | 1.0000 |
| 7 | 0.390395 | 0.152408 | 7.62042 | 77.0285 | $\widetilde{\lambda}_k = 0.8663$ | | |

In order to clearly define the relations among the categories of variables in a threedimensional space, the study used Ward's method with a Euclidean distance. The optimum number of clusters was determined based on the first notable increase in agglomeration distance for the subsequent stages of linkage.

Figure 6 shows linkages of categories into classes, where the horizontal line marks the stage when the linking of classes was interrupted.



Figure 6. Hierarchical classification of variable categories performed with the use of Ward's method.

On the basis of this criterion, the dendrogram was cut at the level of 1.42, obtaining four homogenous clusters which can be characterised as follows:

- Group I (IIE3, TIE3) regards articles concerning innovation and technologies in industry and in enterprises;
- Group II (PI3, IIE2, IIE2) regards articles concerning problems of policy and innovation, as well as innovations and technologies connected with industry and enterprises;
- Group III (MS3, PI2, A1, SD3, A2, MS2, SD2) regards publications concerning China and Europe, which address problems connected with at least two aspects of SD, models and systems of green transformation, and also policy and/or innovation;
- Group IV (IIE1, TIE1, MS1, A3, PI1, SD1) regards publications from the world's areas other than China and Europe that refer to green transformation but do not address issues related to sustainable development (SD), technology and innovation in industry and in enterprises, policy and innovation, as well as models and systems of green transformation.

5. Discussion

Sustainable development goals (SDGs) aim at the integration of three aspects of development: social, economic and environmental, one of those being SDG7, which is meant to provide everyone with access to inexpensive, reliable and sustainable modern energy. In order to be able to achieve this aim, it is necessary to improve energy efficiency and speed up the transformation [114]. Green transformation relates to increased effectiveness of using resources of energy, reduced emission of pollutants, reduced impact on the environment, improved efficiency of work and increased ability for sustainable development, allowing for achieving a situation favourable both to the economy and to the environment [95]. In the opinion of Li and Lin [115], technological progress provides the basis for improving energy effectiveness and energy saving while, at the same time, it leads to stimulating green transformation. This view is shared by Söderholm [116], claiming that one of the significant elements of green transformation is the promotion of development based on sustainable technologies, i.e., such models of production and consumption, which have a far less negative effect on the natural environment, including climate change. According to Ngai and Pissarides [117], technological innovation is a constant source of sustainable development of the modern economy and plays a key role in improving the effectiveness of the use of resources and in the modernisation of industrial infrastructure. Innovations can encourage the effectiveness of the use of natural resources both in the traditional and emerging sectors of the economy [118] and, in this way, contribute to the development of a green economy. A completely different viewpoint was presented by Stern and Rydge [25], who suggested that new technologies aimed at reducing the negative impact of the economy

on the environment are unable to speed up significant ecological transformation. This is also confirmed by the current geopolitical situation and the decidedly faster-than-expected increase in interest on the part of consumers in the highly eco-effective technologies caused by the drastic increase in prices of energy and the risk of limited access to its sources.

This was also confirmed by the research results of this study. In the log-linear model describing the occurrence of the models of green transformation in scientific publications, among the dependent variables, there were also found such variables as "technologies". However, it is worth confirming in line with Stern and Rydge [25] that the technological development itself may not be a sufficient factor to decide about the pace of the process of transformation as it is also important in what setting, system, or even environment these technologies are developing. In the more developed economies, one can observe a greater willingness on the part of society to invest in this type of solution, whereas energy transformation in countries that are just catching up with the leaders may, for a variety of reasons, proceed more slowly. Hence the recent popularity of publications on the so-called just transition draws attention to the fact that the process of transition from a high-emission to a zero-emission economy should consider the level of wealth of the local population [119]. This concept is also linked with the term "co-creation", pointing out that energy transformation is the work of not only governments, introducing appropriate regulations; inventors, creating new technological solutions; and entrepreneurs, launching their production, as well as being affected by the local community, directly interested in introducing them into their homes [120].

During the analyses of the progress towards green transformation, researchers also indicate the need to consider the socio-economic context. According to Crespi [8], not only the environmental aims should be reached without harming economic competitiveness, productivity and economic growth, but also the policy framework created in order to promote environmental sustainability should be able to support economic recovery and increased employment.

These considerations highlight the complexity of the discussed issues and the need to prepare a broad analytical and political framework regulating the transition to a green economy through attaining both environmental and economic goals. This is also confirmed by the systematic literature review undertaken in this study and the modelling based on it. It was found that the models of green transformation in the elaborated log-linear model were influenced by the main factors as well as the interactions for the variables "economy" and "environment".

Energy transformation is also increasingly often addressed in the context of the security of energy supply [121], political security [122] and even the sense of security among the inhabitants of individual countries [123]. There is also an increased number of publications that, more than ever before, concentrate on access to energy as one of the factors that guarantee the functioning of entire economies and that are now becoming increasingly common [124–126].

The subject literature also indicates the necessity of conducting research and monitoring green transformation at the level of both countries and regions [127,128]. According to Barbier [129], this applies in particular to rural areas, where the facilitating wider popularisation and reception of renewable energy and the improved technologies related to energy effectiveness depends on overcoming key obstacles such as lack of long-term systems of financing, weak participation of the private sector, inadequate institutional infrastructure, lack of coordination between local and national governments and weak purchasing power of rural communities.

The scope of the conducted research is very wide—from fairly general terminological divagations directed at, among others, recognising the barriers slowing down the process of green transformation [130,131] to more detailed studies aimed at technological solutions, which can speed up energy transformation such as those related to [132].

In the coming years, one can expect a significant increase in the number of publications on green transition and energy transition. Due to the current energy crisis and the need to

17 of 22

find urgent systemic solutions to accelerate energy transition, this will become even more than now the subject of research on the part of scientists and entrepreneurs representing different scientific disciplines and economic sectors.

6. Conclusions

Alongside the ever more visible problems of the global consumption of resources and the pollution of the environment, the green transformation has become one of the requirements in the world's development, as well as a key concept in research on climate change and sustainable development. This is based on the understanding that the challenges resulting from climate change are so far-reaching that the current graduał approach to these challenges is insufficient. The subject is also becoming increasingly popular in scientific literature and in political discourse, yet there is still a shortage of empirical studies that examine the functioning of this concept and its linkages with other areas. This article, via the analysis of scientific articles and using methods of multivariate statistical analysis, has contributed to filling this gap.

As demonstrated by the authors' research, over the last decade, there were numerous publications devoted to the matter of green transformation, which confirms the growing knowledge and interest in global changes in the environment, such as climate change, loss of biodiversity and energy security. At the same time, this increased interest has led to the emergence of various conceptualisations of this idea.

The conducted systematic literature review, and the statistical analyses (log-linear analysis) based on it, suggest that studies concerning models of green transformation are conducted both at national and regional levels and are usually connected with the economy and the environment. Employing the multivariate correspondence analysis allowed for distinguishing a group of scientific articles that are similar due to the analysed issues. It was found that publications regarding China and Europe most frequently address problems related to at least two aspects of sustainable development in different configurations encompassing economic, environmental and social problems, as well as models and systems of green transformation, policy and innovation. Although publications concerning other geographic regions also include the concept of green transformation, they do not address issues related to sustainable development, technologies and innovation in industry and in enterprises, policy and innovation, as well as models and systems of green transformation.

This article, being a continuation of the earlier studies by the authors on the subject of green energy transition [1], provides a very detailed review of the subject literature. Its purpose was to recognise the existing directions of research and indicate new potentially emerging fields of study. In forthcoming articles, the authors are planning to carry out a review of publications that will appear in bibliographic databases in the next few years and already during the current energy crisis. A limitation of this kind of consideration is the necessity to establish a certain framework for the search for keywords that could be used to conduct a systematic literature review. In this case, the knowledge and experience of the authors may prove to be both an asset and a factor limiting the scope of this search.

Author Contributions: Conceptualisation, I.B. and K.C.; data curation, I.B. and K.C.; formal analysis, I.B. and K.C.; funding acquisition, I.B. and K.C.; investigation, I.B. and K.C.; methodology, I.B. and K.C.; project administration, I.B. and K.C.; resources, I.B. and K.C.; software, I.B. and K.C.; supervision, Iwona Bak and Katarzyna Cheba; Validation, Iwona Bak and Katarzyna Cheba; Validation, Iwona Bak and Katarzyna Cheba; Validation, I.B. and K.C.; writing—review and editing, I.B. and K.C. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

Data Availability Statement: Not applicable.

Conflicts of Interest: The authors declare no conflict of interest.

References

- 1. Bak, I.; Cheba, K. Green Energy. Meta-Analysis of the Research Results; Springer: Cham, Switzerland, 2022.
- 2. Butzer, K.W. Collapse, environment, and society. Proc. Natl. Acad. Sci. USA 2012, 109, 3632–3639. [CrossRef] [PubMed]
- 3. Feola, G. Societal transformation in response to global environmental change: A review of emerging concepts. *Ambio* **2012**, *44*, 376–390. [CrossRef]
- Beddoe, R.R.; Costanza, J.; Farley, E.; Garza, J.; Kent, I.; Kubiszewski, L.; Martinez, T.; McCowen, T.; Murphy, K.; Myers, N.; et al. Overcoming systemic roadblocks to sustainability: The evolutionary redesign of worldviews, institutions, and technologies. *Proc. Natl. Acad. Sci. USA* 2009, 106, 2483–2489. [CrossRef] [PubMed]
- 5. Scoones, I.; Newell, P.; Leach, M. The politics of green transformations. In *The Politics of Green Transformations*; Scoones, I., Leach, M., Newell, P., Eds.; Routledge: London, UK, 2015; pp. 1–24.
- 6. Amundsen, H.; Hermansen, E. Green transformation is a boundary object: An analysis of conceptualisation of transformation in Norwegian primary industries. *Environ. Plan. E Nat. Space* 2021, *4*, 864–885. [CrossRef]
- 7. Brown, D.; McGranahan, G. The urban informal economy, local inclusion and achieving a global green transformation. *Habitat Int.* **2016**, *53*, 97–105. [CrossRef]
- Crespi, F. Policy complexity and the green transformation of the economies as an emergent system property. *Environ. Econ. Policy* Stud. 2016, 18, 143–157. [CrossRef]
- Esposito, M.; Haider, A.; Samaan, D.; Semmler, W. Enhancing job creation through green transformation. *Green Ind. Policy* 2017, 151, 2013.
- 10. Urban, F.; Wang, Y.; Geall, S. Prospects, politics, and practices of solar energy innovation in China. *J. Environ. Dev.* **2018**, 27, 74–98. [CrossRef]
- 11. Stirling, A. Emancipating transformations: From controlling 'the transition' to culturing plural radical progress. In *The Politics of Green Transformations*; Scoones, I., Leach, M., Newell, P., Eds.; Routledge: London, UK, 2015.
- 12. European Union. *Proposal for a Council Recommendation on Ensuring a Fair Transition towards Climate Neutrality;* COM(2021) 801 Final; European Commission: Strasbourg, France, 2021.
- 13. Georgeson, L.; Maslin, M.; Poessinouw, M. The global green economy: A review of concepts, definitions, measurement methodologies and their interactions. *Geo Geogr. Environ.* **2017**, *4*, e00036. [CrossRef]
- 14. Eakin, H.; Shelton, R.E.; Siqueiros-Garcia, J.M.; Charli-Joseph, L.; Manuel-Navarrete, D. Loss and social-ecological transformation. *Ecol. Soc.* **2019**, *24*, 15. [CrossRef]
- 15. Baran, B. Support for renewable energy in Germany as an example of effective public policy. *Oeconomia Copernic.* **2015**, *6*, 143–158. [CrossRef]
- 16. Newell, P. The politics of green transformations in capitalism. In *The Politics of Green Transformations*; Scoones, I., Leach, M., Newell, P., Eds.; Routledge: London, UK, 2015; pp. 86–103.
- 17. Schmitz, H.; Becker, B. From Sustainable Development to the Green Transformation—A Rough Guide; IDS Briefing Paper; IDS: Brighton, UK, 2013.
- 18. WBGU. World in Transition. A Social Contract for Sustainability. Summary for Policy-Makers; WBGU: Berlin, Germany, 2011.
- 19. Lederer, M.; Wallbott, L.; Bauer, S. Tracing sustainability transformations and drivers of Green Economy approaches in the Global South. *J. Environ. Dev.* **2018**, *27*, 3–25. [CrossRef]
- 20. Zhai, X.; An, Y. The relationship between technological innovation and green transformation efficiency in China: An empirical analysis using spatial panel data. *Technol. Soc.* **2021**, *64*, 101498. [CrossRef]
- 21. Us, Y.O.; Pimonenko, T.V.; Tambovceva, T.; Segers, J.P. Green transformations in the healthcare system: The COVID-19 impact. *Health Econ. Manag. Rev.* 2020, 1, 48–59. [CrossRef]
- 22. Gu, S.; Xie, M.; Zhang, X. The Driving Force of Green Transformation and Development. In *Green Transformation and Development*; Palgrave Macmillan: Singapore, 2019; pp. 77–94.
- 23. Urban, F.; Nordensvärd, J. Low carbon development: Origins, concepts and key issues. In *Low Carbon Development*; Routledge: London, UK, 2013; pp. 25–44.
- 24. Du, K.; Cheng, Y.; Yao, X. Environmental regulation, green technology innovation, and industrial structure upgrading: The road to the green transformation of Chinese cities. *Energy Econ.* **2021**, *98*, 105247. [CrossRef]
- 25. Stern, N.; Rydge, J. The new energy-industrial revolution and international agreement on climate change. *Econ. Energy Environ. Policy* **2012**, *1*, 101–120. [CrossRef]
- Folke, C.; Carpenter, S.R.; Walker, B.; Scheffer, M.; Chapin, T.; Rockström, J. Resilience thinking: Integrating resilience, adaptability and transformability. *Ecol. Soc.* 2010, 15, 20. [CrossRef]
- 27. Gillard, R.; Gouldson, A.; Paavola, J.; Van Alstine, J. Transformational responses to climate change: Beyond a systemsperspective of social change in mitigation and adaptation. *Wiley Interdiscip. Rev. Clim. Chang.* **2016**, *7*, 251–265. [CrossRef]
- 28. Kates, R.W.; Travis, W.R.; Wilbanks, T.J. Transformational adaptation when incremental adaptations to climate change are insufficient. *Proc. Natl. Acad. Sci. USA* 2012, 109, 7156–7161. [CrossRef]
- 29. O'Brien, K. Global environmental change II: From adaptation to deliberate transformation. *Prog. Hum. Geogr.* **2012**, *36*, 667–676. [CrossRef]
- 30. Pelling, M.; Manuel-Navarrete, D. From resilience to transformation: The adaptive cycle in two Mexican urban centers. *Ecol. Soc.* **2011**, *16*, 11. [CrossRef]

- 31. Pelling, M.; O'Brien, K.; Matyas, D. Adaptation and transformation. Clim. Chang. 2015, 133, 113–127. [CrossRef]
- 32. Hoad, D. Reflections on small island states and the international climate change negotiations (COP21, Paris, 2015). *Isl. Stud. J.* **2015**, *10*, 259–262. [CrossRef]
- Singh, S.K.; Del Giudice, M.; Chierici, R.; Graziano, D. Green innovation and environmental performance: The role of green transformational leadership and green human resource management. *Technol. Forecast. Soc. Chang.* 2020, 150, 119762. [CrossRef]
- 34. Chen, Y.S.; Chang, C.H. The determinants of green product development performance: Green dynamic capabilities, green transformational leadership, and green creativity. *J. Bus. Ethics* **2013**, *116*, 107–119. [CrossRef]
- Zhang, Y.; Zhang, N.; Tang, Z.R.; Xu, Y.J. Identification of Bi₂WO₆ as a highly selective visible-light photocatalyst toward oxidation of glycerol to dihydroxyacetone in water. *Chem. Sci.* 2013, *4*, 1820–1824. [CrossRef]
- 36. Hess, D.J. Sustainability transitions: A political coalition perspective. Res. Policy 2014, 43, 278–283. [CrossRef]
- Karimi, B.; Biglari, A.; Clark, J.H.; Budarin, V. Green, Transition-Metal-Free Aerobic Oxidation of Alcohols Using a Highly Durable Supported Organocatalyst. *Angew. Chem. Int. Ed.* 2007, 46, 7210–7213. [CrossRef]
- Seidel, S.; Recker, J.; Vom Brocke, J. Sensemaking and sustainable practicing: Functional affordances of information systems in green transformations. *MIS Q.* 2013, 37, 1275–1299. [CrossRef]
- Cai, W.; Lai, K.H.; Liu, C.; Wei, F.; Ma, M.; Jia, S.; Lv, L. Promoting sustainability of manufacturing industry through the lean energy-saving and emission-reduction strategy. *Sci. Total Environ.* 2019, 665, 23–32. [CrossRef]
- Ouyang, X.; Li, Q.; Du, K. How does environmental regulation promote technological innovations in the industrial sector? Evidence from Chinese provincial panel data. *Energy Policy* 2020, 139, 111310. [CrossRef]
- Mittal, S.; Dhar, R.L. Effect of green transformational leadership on green creativity: A study of tourist hotels. *Tour. Manag.* 2016, 57, 118–127. [CrossRef]
- 42. Booth, A.; Sutton, A.; Papaioannou, D. Systematic Approaches to a Successful Literature Review; Sage: Los Angeles, CA, USA, 2012.
- 43. Czakon, W. Metodyka systematycznego przeglądu literatury. Przegląd Organ. 2011, 3, 57–61. [CrossRef]
- 44. Tranfield, D.; Denyer, D. Towards a Methodology for Developing Evidence-Informed Management Knowledge by Means of Systematic Review. *Br. Acad. Manag.* 2003, *14*, 207–222. [CrossRef]
- 45. Pautasso, M. Ten Simple Rules for Writing a Literature Review. PLoS Comput. Biol. 2013, 9, e1003149. [CrossRef]
- 46. Winchester, C.L.; Salji, M. Writing a literature review. J. Clin. Urol. 2016, 9, 308–312. [CrossRef]
- Maggio, L.A.; Sewell, J.L.; Artino, A.R. The Literature Review: A Foundation for High-Quality Medical Education Research. J. Grad. Med. Educ. 2016, 8, 297–303. [CrossRef]
- 48. Kovacova, M.; Kliestik, T.; Valaskova, K.; Durana, P.; Juhaszova, Z. Systematic review of variables applied in bankruptcy prediction models of Visegrad group countries. *Oeconomia Copernic.* **2019**, *10*, 743–772. [CrossRef]
- Tiensuwan, M.; Yimprayoon, P.; Lenbury, Y. Application of log-linear models to cancer patients: A case study of data from the National Cancer Institute. *Southeast Asian J. Trop. Med. Public Health* 2005, *36*, 1283.
- Oğüş, E. Comparison of Log-Linear Analysis and Correspondence Analysis in Two-Way Contingency Tables: A Medical Application. *Balk. Med. J.* 2011, 28, 143–147.
- 51. Fatin, A.D.; Pavlenko, E.Y.; Poltavtseva, M.A. A survey of mathematical methods for security analysis of cyberphysical systems. *Autom. Control. Comput. Sci.* 2020, 54, 983–987. [CrossRef]
- Ahmad, F.; Chand, S. Log Linear Models for Religious and Social Factors affecting the practice of Family Planning Methods in Lahore, Pakistan. Pak. J. Stat. Oper. Res. 2006, 2, 35–44. [CrossRef]
- 53. Adarabioyo, M.I. Application of log-linear model to determinants of child mortality in Nigeria. *Int. J. Sci. Res. Innov. Technol* **2014**, 1, 34–44.
- 54. Salamaga, M. Wykorzystanie analizy log-liniowej do wyboru czynników opisujących sytuację ekonomiczną gospodarstw domowych. *Przegląd Stat.* **2008**, *55*, 40–51.
- 55. Zioło, M.; Bak, I.; Filipiak, B.Z.; Spoz, A. In search of a financial model for a sustainable economy. *Technol. Econ. Dev. Econ.* 2022, 28, 920–947. [CrossRef]
- Skinner, C.J.; Shlomo, N. Assessing identification risk in survey microdata using log-linear models. J. Am. Stat. Assoc. 2008, 103, 989–1001. [CrossRef]
- 57. Lee, M.-J. Finding correct elasticities in log-linear and exponential models allowing heteroskedasticity. *Stud. Nonlinear Dyn. Econom.* **2020**, *25*, 81–91. [CrossRef]
- 58. Grybaitė, V.; Stankevičienė, J. An empirical analysis of factors affecting sharing economy growth. *Oeconomia Copernic.* 2018, 9, 635–654. [CrossRef]
- Kudlats, J.; Money, A.; Hair, J.F., Jr. Correspondence analysis: A promising technique to interpret qualitative data in family business research. J. Fam. Bus. Strategy 2014, 5, 30–40. [CrossRef]
- Ayele, D.; Zewotir, T.; Mwambi, H. Multiple correspondence analysis as a tool for analysis of large health surveys in African settings. *Afr. Health Sci.* 2014, 14, 1036–1045. [CrossRef]
- Belas, J.; Gavurova, B.; Čepel, M.; Kubák, M. Evaluation of economic potential of business environment development by comparing sector differences: Perspective of SMEs in the Czech Republic and Slovakia. *Oeconomia Copernic.* 2020, 11, 135–159. [CrossRef]
- 62. Greenacre, M. Theory Applications of Correspondence Analysis; Academic Press: London, UK, 1984.

- 63. Greenacre, M. Multiple and joint correspondence analysis. In *Correspondence Analysis in Social Sciences. Recent Developments and Applications*; Greenacre, M., Blasius, J., Eds.; Academic Press: San Diego, CA, USA, 1994; pp. 141–161.
- 64. Lebart, L.; Morineau, A.; Warwick, K.M. Multivariate Descriptive Statistical Analysis. Correspondence Analysis and Related Techniques for Large Matrices; John Wiley & Sons, Inc.: New York, NY, USA, 1984.
- 65. Greenacre, M. Correspondence Analysis in Practice; Academic Press: London, UK, 1993.
- Colasante, A.; D'Adamo, I.; Morone, P. What drives the solar energy transition? The effect of policies, incentives and behavior in a cross-country comparison. *Energy Res. Soc. Sci.* 2022, 85, 102405. [CrossRef]
- Schittekatte, T.; Meeus, L.; Jamasb, T.; Llorca, M. Regulatory experimentation in energy: Three pioneer countries and lessons for the green transition. *Energy Policy* 2021, 156, 112382. [CrossRef]
- 68. Terzi, A. Crafting an effective narrative on the green transition. Energy Policy 2020, 147, 111883. [CrossRef]
- 69. Kylili, A.; Thabit, Q.; Nassour, A.; Fokaides, P.A. Adoption of a holistic framework for innovative sustainable renewable energy development: A case study. *Energy Sources Part A Recovery Util. Environ. Eff.* **2021**, 1–21. [CrossRef]
- Li, H.; Li, B. The threshold effect of environmental regulation on the green transition of the industrial economy in China. *Econ. Res. Ekon. Istraživanja* 2019, 32, 3128–3143. [CrossRef]
- Tan, H.; Thurbon, E.; Kim, S.Y.; Mathews, J.A. Overcoming incumbent resistance to the clean energy shift: How local governments act as change agents in coal power station closures in China. *Energy Policy* 2021, 149, 112058. [CrossRef]
- Yuan, R.; Li, C.; Li, N.; Khan, M.A.; Sun, X.; Khaliq, N. Can mixed-ownership reform drive the green transformation of SOEs? *Energies* 2021, 14, 2964. [CrossRef]
- 73. Grant, R.; Carmody, P.; Murphy, J.T. A green transition in South Africa? Sociotechnical experimentation in the Atlantis special economic zone. *J. Mod. Afr. Stud.* 2020, *58*, 189–211. [CrossRef]
- 74. Yu, L.; Wang, S.; Lu, Q.; Feng, G. Sensitivity analysis of existing residential building energy consumption influencing factors in cold regions. *Procedia Eng.* **2016**, *146*, 196–203. [CrossRef]
- Matsuo, T.; Schmidt, T.S. Managing tradeoffs in green industrial policies: The role of renewable energy policy design. *World Dev.* 2019, 122, 11–26. [CrossRef]
- 76. Schmitz, H. Who drives climate-relevant policies in the rising powers? New Political Econ. 2017, 22, 521–540. [CrossRef]
- 77. Kennedy, C.; Zhong, M.; Corfee-Morlot, J. Infrastructure for China's ecologically balanced civilization. *Engineering* **2016**, 2, 414–425. [CrossRef]
- Li, J. Empirical Study on Green Design Transformation of the Old Industrial Buildings. In Proceedings of the 2015 International Conference on Economics, Social Science, Arts, Education and Management Engineering, Xi'an, China, 12–13 December 2015; pp. 255–258.
- Schmitz, H.; Johnson, O.; Altenburg, T. Rent management-the heart of green industrial policy. *IDS Work. Pap.* 2013, 2013, 1–26. [CrossRef]
- Wang, X.; He, F.; Zhang, L.; Chen, L. Energy efficiency of China's iron and steel industry from the perspective of technology heterogeneity. *Energies* 2018, 11, 1247. [CrossRef]
- Yang, J.; Gu, F.; Guo, J. Environmental feasibility of secondary use of electric vehicle lithium-ion batteries in communication base stations. *Resour. Conserv. Recycl.* 2020, 156, 104713. [CrossRef]
- 82. Dunlap, A. Spreading 'green'infrastructural harm: Mapping conflicts and socio-ecological disruptions within the European Union's transnational energy grid. *Globalizations* **2021**, 1–25. [CrossRef]
- 83. Huang, G.C.L.; Chen, R.Y. Injustices in phasing out nuclear power? Exploring limited public participation and transparency in Taiwan's transition away from nuclear energy. *Energy Res. Soc. Sci.* **2021**, *71*, 101808. [CrossRef]
- 84. Deberdt, R.; Le Billon, P. Conflict minerals and battery materials supply chains: A mapping review of responsible sourcing initiatives. *Extr. Ind. Soc.* 2021, *8*, 100935. [CrossRef]
- Groppi, D.; Garcia, D.A.; Basso, G.L.; Cumo, F.; De Santoli, L. Analysing economic and environmental sustainability related to the use of battery and hydrogen energy storages for increasing the energy independence of small islands. *Energy Convers. Manag.* 2018, 177, 64–76. [CrossRef]
- 86. Krog, L. How municipalities act under the new paradigm for energy planning. Sustain. Cities Soc. 2019, 47, 101511. [CrossRef]
- Ferrari, G.; Pezzuolo, A.; Nizami, A.S.; Marinello, F. Bibliometric Analysis of Trends in Biomass for Bioenergy Research. *Energies* 2020, 13, 3714. [CrossRef]
- Pallesen, T.; Jacobsen, P.H. Demonstrating a flexible electricity consumer: Keeping sight of sites in a real-world experiment. *Sci. Cult.* 2021, 30, 172–191. [CrossRef]
- 89. Gurwitz, R.; Cohen, R.; Shalish, I. Interaction of light with the ZnO surface: Photon induced oxygen "breathing," oxygen vacancies, persistent photoconductivity, and persistent photovoltage. *J. Appl. Phys.* **2014**, *115*, 033701. [CrossRef]
- 90. Barbier, E.B. Is green rural transformation possible in developing countries? World Dev. 2020, 131, 104955. [CrossRef]
- Wang, Y. China's Transition to Green Development: Process, Challenges and Responsive Measures. *Chin. J. Urban Environ. Stud.* 2020, *8*, 2075005. [CrossRef]
- Bakar, N.N.A.; Guerrero, J.M.C.; Vasquez, J.; Bazmohammadi, N.; Othman, M.; Rasmussen, B.D.; Al-Turki, Y.A. Optimal Configuration and Sizing of Seaport Microgrids including Renewable Energy and Cold Ironing—The Port of Aalborg Case Study. *Energies* 2022, 15, 431. [CrossRef]

- 93. Haukkala, T. A struggle for change—The formation of a green-transition advocacy coalition in Finland. *Environ. Innov. Soc. Transit.* **2018**, 27, 146–156. [CrossRef]
- 94. Yang, Z.; Shao, S.; Yang, L.; Miao, Z. Improvement pathway of energy consumption structure in China's industrial sector: From the perspective of directed technical change. *Energy Econ.* **2018**, *72*, 166–176. [CrossRef]
- Xie, W.; Yan, T.; Xia, S.; Chen, F. Innovation or introduction? The impact of technological progress sources on industrial green transformation of resource-based cities in China. *Front. Energy Res.* 2020, *8*, 598141. [CrossRef]
- 96. Amendolagine, V.; Lema, R.; Rabellotti, R. Green foreign direct investments and the deepening of capabilities for sustainable innovation in multinationals: Insights from renewable energy. *J. Clean. Prod.* **2021**, *310*, 127381. [CrossRef]
- 97. Midttun, A. The greening of European electricity industry: A battle of modernities. Energy Policy 2012, 48, 22–35. [CrossRef]
- Wang, C. Explaining Connotation of Regional Green Transition. In Proceedings of the 2015 International Conference on Industrial Technology and Management Science 2015, Tianjin, China, 27–28 March 2015; pp. 1203–1205.
- 99. Varela-Vázquez, P.; Sánchez-Carreira, M.D.C. Upgrading peripheral wind sectors. *Technol. Anal. Strateg. Manag.* 2016, 28, 1152–1166. [CrossRef]
- Pelorosso, R.; Gobattoni, F.; Ripa, M.N.; Leone, A. Second law of thermodynamics and urban green infrastructure–A knowledge synthesis to address spatial planning strategies. *TeMA J. Land Use Mobil. Environ.* 2018, 11, 27–50.
- 101. Kuhn, B.M. China's commitment to the sustainable development goals: An analysis of push and pull factors and implementation challenges. *Chin. Political Sci. Rev.* 2018, *3*, 359–388. [CrossRef]
- 102. Pegels, A.; Vidican-Auktor, G.; Lütkenhorst, W.; Altenburg, T. Politics of green energy policy. J. Environ. Dev. 2018, 27, 26–45. [CrossRef]
- Lu, Y.; Chang, R.; Chong, D.; Ngiam, M.L.J. Transition towards green facility management: Bridging the knowledge gaps of facilities managers. J. Green Build. 2018, 13, 122–143. [CrossRef]
- 104. MacArthur, J.L.; Hoicka, C.E.; Castleden, H.; Das, R.; Lieu, J. Canada's Green New Deal: Forging the socio-political foundations of climate resilient infrastructure? *Energy Res. Soc. Sci.* 2020, 65, 101442. [CrossRef]
- Cong, R.G.; Caro, D.; Thomsen, M. Is it beneficial to use biogas in the Danish transport sector? An environmental-economic analysis. J. Clean. Prod. 2017, 165, 1025–1035. [CrossRef]
- 106. Li, J.; Wei, W.; Zhen, W.; Guo, Y.; Chen, B. How green transition of energy system impacts China's mercury emissions. *Earth's Future* **2019**, *7*, 1407–1416. [CrossRef]
- 107. Horsbøl, A.; Andersen, P.V. Actors and agency in district heating: Engaging with middle actor perspectives through future workshops. *Energy Res. Soc. Sci.* 2021, *80*, 102200. [CrossRef]
- 108. Fang, S.; Xue, X.; Yin, G.; Fang, H.; Li, J.; Zhang, Y. Evaluation and improvement of technological innovation efficiency of new energy vehicle enterprises in China based on DEA-Tobit model. *Sustainability* **2020**, *12*, 7509. [CrossRef]
- 109. Rasmussen, N.B.; Enevoldsen, P.; Xydis, G. Transformative multivalue business models: A bottom-up perspective on the hydrogen-based green transition for modern wind power cooperatives. *Int. J. Energy Res.* **2020**, *44*, 3990–4007. [CrossRef]
- Lahcen, B.; Brusselaers, J.; Vrancken, K.; Dams, Y.; Da Silva Paes, C.; Eyckmans, J.; Rousseau, S. Green recovery policies for the COVID-19 crisis: Modelling the Impact on the economy and greenhouse gas emissions. *Environ. Resour. Econ.* 2020, 76, 731–750. [CrossRef] [PubMed]
- 111. Lamperti, F.; Dosi, G.; Napoletano, M.; Roventini, A.; Sapio, A. Climate change and green transitions in an agent-based integrated assessment model. *Technol. Forecast. Soc. Chang.* 2020, 153, 119806. [CrossRef]
- 112. D'Adamo, I.; Falcone, P.M.; Huisingh, D.; Morone, P. A circular economy model based on biomethane: What are the opportunities for the municipality of Rome and beyond? *Renew. Energy* **2021**, *163*, 1660–1672. [CrossRef]
- 113. Wang, J.; Cao, H. Improving competitive strategic decisions of Chinese coal companies toward green transformation: A hybrid multi-criteria decision-making model. *Resour. Policy* **2022**, *75*, 102483. [CrossRef]
- 114. Adom, P.K.; Adams, S. Technical fossil fuel energy efficiency (TFFEE) and debt-finance government expenditure nexus in Africa. *J. Clean. Prod.* **2020**, 271, 122670. [CrossRef]
- 115. Li, K.; Lin, B. How to promote energy efficiency through technological progress in China? Energy 2018, 143, 812–821. [CrossRef]
- 116. Söderholm, P. The green economy transition: The challenges of technological change for sustainability. *Sustain. Earth* **2020**, *3*, 6. [CrossRef]
- 117. Ngai, L.R.; Pissarides, C.A. Structural change in a multisector model of growth. Am. Econ. Rev. 2007, 97, 429-443. [CrossRef]
- Miao, C.; Fang, D.; Sun, L.; Luo, Q.; Yu, Q. Driving effect of technology innovation on energy utilization efficiency in strategic emerging industries. J. Clean. Prod. 2018, 170, 1177–1184. [CrossRef]
- 119. Heffron, R.J. What is the "just transition"? In *Achieving a Just Transition to a Low-Carbon Economy*; Palgrave Macmillan: Cham, Switzerland, 2021; pp. 9–19.
- Sillak, S.; Borch, K.; Sperling, K. Assessing co-creation in strategic planning for urban energy transitions. *Energy Res. Soc. Sci.* 2021, 74, 101952. [CrossRef]
- 121. Scipioni, A.; Manzardo, A.; Ren, J. Hydrogen Economy: Supply Chain, Life Cycle Analysis and Energy Transition for Sustainability; Academic Press: Cambridge, MA, USA, 2017.
- Park, J.; Kim, B. An analysis of South Korea's energy transition policy with regards to offshore wind power development. *Renew. Sustain. Energy Rev.* 2019, 109, 71–84. [CrossRef]

- 123. Carley, S.; Konisky, D.M. The justice and equity implications of the clean energy transition. *Nat. Energy* **2020**, *5*, 569–577. [CrossRef]
- 124. Farhat, H.; Salvini, C. Novel Gas Turbine Challenges to Support the Clean Energy Transition. Energies 2022, 15, 5474. [CrossRef]
- 125. Kim, Y.J.; Soh, M.; Cho, S.H. Identifying optimal financial budget distributions for the low-carbon energy transition between emerging and developed countries. *Appl. Energy* **2022**, *326*, 119967. [CrossRef]
- 126. Doran, R.; Böhm, G.; Pfister, H.R.; Hanss, D. Mapping perceptions of energy transition pathways: Ascribed motives and effectiveness. *Current Psychology* 2022, 1–13. [CrossRef]
- 127. Tsonkov, N. Challenges and opportunities for green economy transformation of the Gabrovo District. SHS Web Conf. 2021, 120, 01007. [CrossRef]
- 128. Morales, D.; Sariego-Kluge, L. Regional state innovation in peripheral regions: Enabling Lapland's green policies. *Reg. Stud.* 2021, *8*, 54–64. [CrossRef]
- 129. Barbier, E.B. Greening the post-pandemic recovery in the G20. Environ. Resour. Econ. 2020, 76, 685–703. [CrossRef] [PubMed]
- Mao, W.; Wang, W.; Sun, H. Driving patterns of industrial green transformation: A multiple regions case learning from China. *Sci. Total Environ.* 2019, 697, 134134. [CrossRef] [PubMed]
- 131. Ul Ain, Q.; Shafique, O. Development and validation of a questionnaire to identify the barriers to adopting green transformation. *Compet. Educ. Res. J.* **2022**, *3*, 77–103.
- 132. Liu, J.; Li, J.; Wang, J.; Uddin, M.M.; Zhang, B. Research on the Application of Blockchain Technology in Coal Supply Chain Finance. *Sustainability* **2022**, *14*, 10099. [CrossRef]

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.